Author’s response to reviews

**Title:** The Low Indexes of Metabolism Intervention Trial (LIMIT): Design and baseline data of a randomized controlled clinical trial to evaluate how alerting primary care teams to low metabolic values, could affect the health of patients aged 75 or older

**Authors:**

Nir Tsabar (Tsabar.Nir@Gmail.Com;nir.tabar@clalit.org.il)

Yan Press (yanpr@clalit.org.il)

Johanna Rotman (hannekero@clalit.org.il)

Bracha Klein (kleinbr@clalit.org.il)

Yonatan Grossman (yonatangr@clalit.org.il)

Maya Vainshtein-Tal (may@clalit.org.il)

Sophia Eilat-Tsanani (eilat@clalit.org.il)

**Version:** 4  **Date:** 16 Dec 2017

**Author’s response to reviews:**

As you requested, we have made all necessary changes in our manuscript.